Author's response to reviews

Title: A Decision Aid for COPD patients considering inhaled steroid therapy: development and before and after pilot testing

Authors:

Elie A Akl (elieakl@buffalo.edu)
Brydon JB Grant (grant@buffalo.edu)
Gordon H Guyatt (guyatt@mcmaster.ca)
Victor M Montori (Montori.Victor@mayo.edu)
Holger J Schunemann (hjs@buffalo.edu)

Version: 5 Date: 14 May 2007

Author's response to reviews: see over
May 14, 2007

RE: “A Decision Aid for COPD patients considering inhaled steroid therapy: development and before and after pilot testing”

Dear Dr. Saltman,

We have addressed the formatting requirements. We are looking forward to the publishing of the manuscript.

This manuscript has not been published in another journal and is not under consideration by any other peer-reviewed media. It presents original work in which all the authors listed on the manuscript have contributed sufficiently to the project to be included as authors.

With kind regards,

Elie A. Akl, MD, MPH
Assistant Professor of Medicine
Department of Medicine, University at Buffalo